Author’s response to reviews

Title: Sero-Prevalence of Hepatitis B Virus and Associated Factors among Pregnant Women in Gambella Hospital, South Western Ethiopia: Facility based cross-sectional study

Authors:
Abayneh tanga (abaynehtun@yahoo.com)
Misanew Teshome (misanew08@gmail.com)
Desta Hiko (destahika@gmail.com)
Chaltu Fikru (fikiruc@yahoo.com)
Gemechu Jilo (gemechukejela86@gmail.com)

Version: 2 Date: 22 Dec 2018

Author’s response to reviews:
All comments given by reviewers are corrected and highlighted:

-abstract
-back ground....reference
-method ....removing sampling formula
-result ....making one table in to two
-limitation under discussion section